Title: Early Identification of Women at Risk of Postpartum Depression using the Edinburgh Postnatal Depression Scale (EPDS) in a sample of Lebanese Women

Authors:

Charline El-Hachem (charlineeh@gmail.com)
Jihane ROHAYYEM (jihanerohayem@yahoo.com)
Rami BOU KHALIL (ramiboukhalil@hotmail.com)
Sami RICHA (samiric@idm.net.lb)
Assaad KESSEROUANI (assaad.kesserouani@hdf.usj.edu.lb)
Rima GEMAYEL (rima.gemayel@hdf.usj.edu.lb)
Norma AOUAD (norma.aouad@hdf.usj.edu.lb)
Najat HATAB (najat.hatab@hdf.usj.edu.lb)
Eliane Zaccak (eliane.zakkak@hdf.usj.edu.lb)
Nancy YAGHI (nancy.yaghi@hotmail.com)
Salimé SALAMEH (salisalameh@hotmail.com)
Elie ATTIEH (elie.attieh@hdf.usj.edu.lb)

Version: 4  Date: 16 August 2014

Author’s response to reviews:

Dear Editor,

My co-authors and I are pleased to submit to “BMC Psychiatry” a revised version of our manuscript entitled: “Early Identification of Women at Risk of Postpartum Depression using the Edinburgh Postnatal Depression Scale (EPDS) in a sample of Lebanese Women”.

We have taken into consideration the three editorial requests and included an author’s contribution and a conclusion section at the end of the manuscript and inserted the tables after the reference list.

My co-authors and I declare having no conflict of interest of any kind behind this work. Data contained in this manuscript have not been published or considered for publication elsewhere. Finally, we have no acknowledgments to declare.

Best regards

Rami Bou Khalil, M.D.